Author's response to reviews

Title: Charcot osteoarthropathy in type 2 diabetes persons presenting to specialist diabetes clinic at a tertiary care hospital.

Authors:

Bilal Bin Younis (bilalbin@gmail.com)
Adeela Shahid (dradeelas@hotmail.com)
Rozina Arshad (rozina979@gmail.com)
Saima Khurshid (dr.cyma@yahoo.com)
Junaid Masood (Dph_junaidmasood@hotmail.com)

Version: 2
Date: 12 December 2014

Author's response to reviews:

Thank you very much for consideration, in response of your email I have revised the manuscript including

(1) Addition of Title Page
(2) Addition of Acknowledgments
(3) Addition of Tables
(4) Addition of Line Numbering